June 11, 2023

Dear editor,

Please find attached files of revised manuscript in word format

**Title:** Nasopharyngeal carcinoma with synchronous breast metastasis: A case report and literature review

**Name of Journal:** World Journal of Clinical Cases

**Manuscript NO:** 85606

First of all, thank you for your careful guidance of this article. Revision has been made according to the suggestions of the reviewer:

**Reviewer: 06143351**

This study reports an interesting case of nasopharyngeal carcinoma with synchronous breast metastasis. The first pathological biopsy report of the case suggested triple-negative breast invasive carcinoma. This case report suggests that breast specialists should carefully rule out secondary breast cancers when diagnosing and treating breast masses. Interesting case, and well discussed. No specific comments to the authors.

**Reviewer: 06074583**

In this study, the authors present the clinical characteristics, treatment strategies, and follow-up information for a patient with newly diagnosed stage IV NPC who developed synchronous breast metastasis. The case is interesting and well described. The manuscript is well written. The reviewer recommends to accept this case report after a minor editing.

After receiving the comments, we read the article carefully and found some small loopholes in the language of the article and made modifications and new language proofs are also provided.

According to the suggestion, we changed the relevant description.

Thank you again for publishing our manuscript in the World Journal of Clinical Cases.
Sincerely Yours,

Dong-Mei Li